Author’s response to reviews

Title: Bisphosphonate-Induced Orbital Inflammation in a Patient on Chronic Immunosuppressive Therapy

Authors:

Isabella Herrera (iherrera0622@gmail.com)
Yong Kam (ykam@kumc.edu)
Thomas Whittaker (twhittaker@kumc.edu)
Mary Champion (mpham@kumc.edu)
Radwan Ajlan (radwanajlan@gmail.com)

Version: 1 Date: 14 Jan 2019

Author’s response to reviews:

January 14, 2018

Editorial Office
BMC Ophthalmology

Bisphosphonate-Induced Orbital Inflammation in a Patient on Chronic Immunosuppressive Therapy

Dear editor and reviewers,

Thank you for providing us with this opportunity to address the editor’s and reviewers’ comments and to resubmit our revised manuscript for consideration for publication in BMC Ophthalmology. In our revisions, we have carefully considered the editor’s and reviewers’ comments, and we have included a point-by-point response to each of the comments below.

Alejandro Fonollosa Calduch (Reviewer 1): Thank you for your comment. We agree that orbital inflammation secondary to systemic lupus erythematosus should be included in the differential based on this patient’s history. The manuscript was revised to include this differential in the Case Presentation section, line 23, page 4.

Ariel Schlaen, MD (Reviewer 2): Thank you for your comments. We have reviewed the patient’s chart again and the patient was actually on a 75mg (1mg/kg) dose of cyclosporine when she presented to our clinic. We have revised the manuscript to reflect this change in the Case

...
Presentation section, lines 10, page 4. We have also edited the typo of “lacrimal” and has been changed to the correct spelling in the Discussion section, line 18, page 5.

Thank you for considering this important and interesting case report for publication in BMC Ophthalmology. We look forward to your decision on publication.

Sincerely,

Radwan Ajlan, MBBC, FRCS(C), FICO, DABO

Assistant professor
Retina and vitreous
Department of ophthalmology
University of Kansas School of Medicine